Author's response to reviews

Title: Sentinel surveillance for human enterovirus 71 in Sarawak, Malaysia: Lessons from the first 7 years.

Authors:

Yuwana Podin (ypodin@yahoo.com)
Edna LM Gias (amberr74@yahoo.com)
Flora Ong (flora.ong@sarawak.health.gov.my)
Yee-Wei Leong (leongyeewei@hotmail.com)
Siew-Fung Yee (siewfungyee@yahoo.co.uk)
Mohd Apandi Yusuf (apandiyusof@yahoo.com)
David Perera (davidperera@yahoo.com)
Bibiana Teo (bibianateo@yahoo.com)
Thian-Yew Wee (thian_57@hotmail.com)
Sik-Chi Yao (sikchi@pc.jaring.my)
Sik-King Yao (yaosk@health.gov.my)
Andrew Kiyu (andrew.kiyu@sarawak.health.gov.my)
Mohd Taha Arif (mtaha@time.net.my)
Mary Jane Cardosa (janecardosa@yahoo.co.uk)

Version: 7 Date: 28 June 2006

Author's response to reviews:

I have made the formatting revisions requested:

1. Included references to the additional files in the text.
2. Added information about the additional files at the end of the manuscript after the tables.
3. Removed the url to APNET website from the text and put it into the reference list.
4. Fixed a slight misalignment which appeared in the figures 2 and 3 (the tiff files)

Additionally I have corrected a couple of typographical errors - eg extra spaces where there should not be any and so on.

The bars in figure one are now fine. The colours work well.

Thank you for very very helpful and accommodating.